Author's response to reviews

Title: End of life care in sub-Saharan Africa: a systematic review of the qualitative literature

Authors:

Marjolein Gysels (marjolein.gysels@cresib.cat)
Christopher Pell (christopher.pell@cresib.cat)
Lianne Straus (liannestraus@hotmail.com)
Robert Pool (Robert.pool@cresib.cat)

Version: 5 Date: 25 February 2011

Author's response to reviews:

The major and minor revisions requested have been made.
The title case has been changed, as have the names of the additional files.
The figure has been re-formated and the article checked for typographical errors (minor changes have been made).
Track changes and highlighting has been removed, as have the vertical lines from the tables.